Author’s response to reviews

Title: Serum 25-hydroxyvitamin D3 levels and Vitamin D Receptor variants in melanoma patients from the mediterranean area of Barcelona

Authors:

Zighereda Ogbah (zigheog@gmail.com)
Laura Visa (lvisa@clinic.ub.es)
Celia Badenas (cbadenas@clinic.ub.es)
José Ríos (jrios@uab.es)
Joan-Anton Puig-Butille (jantonpuig@gmail.com)
Nuria Bonifaci (nbonifaci@iconcologia.net)
Elisabet Guinò (eguino@iconcologia.net)
Josep Maria Augè (jmauge@clinic.ub.es)
Isabel Kolm (ikolm@yahoo.com)
Cristina Carrera (ccarrera@clinic.ub.es)
Miquel Angel Pujana (mapujana@iconcologia.net)
Josep Malvehy (jmalvehy@clinic.ub.es)
Susana Puig (spuig@clinic.ub.es)

Version: 3 Date: 26 July 2012

Author’s response to reviews:

Dear Editor,

thank for you revision. In the Methods section, study design paragraph (page 5-beginning page 6), we have reported that all patients that participated in our studies gave a written informed consent for their participation.

IN TEXT

"Both studies were approved by the Ethics Committee of the Hospital Clinic of Barcelona and written informed consent was obtained from all participants in the study".

Kind regards